Author's response to reviews

Title: PLAYgrounds: Effect of a PE playground program in primary schools on PA levels during recess in 6 to 12 year old children. Design of a prospective controlled trial

Authors:

Mirka Janssen (m.janssen@hva.nl)
Huub M Toussaint (h.toussaint@hva.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)
Evert ALM Verhagen (e.verhagen@vumc.nl)

Version: 4 Date: 13 April 2011

Author's response to reviews:

I checked the manuscript carefully for typing and content errors.

In the author list I changed the list of authors in the manuscript to the right style and order.

I checked all references and made sure the references are in numerical order in the text. Reference 22 was in the correct position, but not in the text. Thank you for indicating this reference.